



Article Morphological Characteristics and Prevention of Tooth Enamel Demineralization during Orthodontic Treatment with Non-Removable Appliances

Gagik Khachatryan¹, Marina Markaryan^{2,*}, Izabella Vardanyan³, Mikayel Manrikyan³ and Gayane Manrikyan²

- ¹ Department of Dental and Pharmacological Professional Education, Yerevan State Medical University (YSMU), Koryun Str. 2, Yerevan 0002, Armenia
- ² Department of Therapeutic Dentistry, Yerevan State Medical University (YSMU), Koryun Str. 2, Yerevan 0002, Armenia
- ³ Department of Pediatric Dentistry and Orthodontics, Yerevan State Medical University (YSMU), Koryun Str. 2, Yerevan 0002, Armenia
- * Correspondence: marmiga@mail.ru; Tel.:+374-91407257

Abstract: Despite the large number of studies on the effect of braces on teeth, there is no information on the dynamics of the state of the ultrastructure of the hard tissues of teeth during orthodontic treatment. The purpose of this study is to examine the state of the hard tissues of the teeth and carry out preventive measures to reduce the risk of complications in the process of orthodontic treatment using a non-removable device. Methods: For the in vitro study, 68 teeth were randomly divided into group A—no prophylactic treatment, and group B—treated with the fluorine varnish Tiefenfluorid. After 35 days, all the teeth were prepared for microscopic examination. The clinical study included 59 patients aged 12–17 years with orthodontic brackets. The split-mouth technique was used. The areas around the bracket of one-half of the oral cavity were treated with Tiefenfluorid every 6 months during the entire treatment period. The teeth of the second half of the oral cavity served as the control group. The data were processed in the SPSS19 package. Results: In vitro and clinical study results showed a statistically significant difference between the prophylactic and control groups of teeth in favor of the prophylactic group, where the statistical significance was $p \leq 0.01$.

Keywords: enamel demineralization; non-removable appliance; prevention; white spot lesion

1. Introduction

With the enlargement in the number of dentoalveolar anomalies, there has been a corresponding increase in the need for orthodontic treatment by means of various types of non-removable devices. Its use is effective for the treatment of anomalies in the position of individual teeth and their groups, as well as occlusion anomalies. In the course of orthodontic treatment over a prolonged period, various non-removable devices are installed in the patient's oral cavity, which are followed by a change in homeostasis that has a negative effect on the dentoalveolar system's tissues and organs. The mastication function and the physical state of the dentoalveolar system change, and the quality of life of patients undergoing orthodontic treatment decreases [1,2]. The hygienic state of the oral cavity in the presence of a bracket system worsens due to various structural elements of the devices which serve as retention points for the accumulation of soft microbial biofilm, while the susceptibility of teeth to caries increases and inflammation develops in the periodontal tissues [3]. The impairment of the dynamic balance of metabolic processes, an increase in the number of pathogenic and opportunistic microorganisms that occur in the process of orthodontic treatment, and the development of serious complications against this background remain relevant. During orthodontic treatment, enamel demineralization is most often detected in the form of "white" spots but only at the stage of removing the



Citation: Khachatryan, G.; Markaryan, M.; Vardanyan, I.; Manrikyan, M.; Manrikyan, G. Morphological Characteristics and Prevention of Tooth Enamel Demineralization during Orthodontic Treatment with Non-Removable Appliances. Int. J. Environ. Res. Public Health 2023, 20, 540. https://doi.org/ 10.3390/ijerph20010540

Academic Editor: Aida Meto

Received: 25 October 2022 Revised: 22 December 2022 Accepted: 23 December 2022 Published: 29 December 2022



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). orthodontic device [4]. A number of authors have studied the state of tooth enamel during orthodontic treatment on the bracket system. Notably, researchers have provided evidence that enamel damage after the removal of braces was quite significant [5,6]. Similar data were obtained by other authors [1,7].

The first signs of enamel demineralization appear 1 month after fixation of the orthodontic device and manifest as white spots that extend to a depth of 100 microns [8–10], which, in turn, can lead to further spread of the carious process if rational oral hygiene measures are not taken. White spot lesions (WSL) develop rapidly and often become an aesthetic problem for patients even years after the orthodontic appliances have been removed. However, their frequency has been reported to be widely variable, from 2% to 97%, in different epidemiological studies [2,9], which might be explained by the techniques used to detect and characterize them, including visual inspection and photographs [11]. In recent years, fluorescent diagnostics using DIAGNOdent devices (KaVo, Germany) for detecting initial caries has become more widespread [12,13].

When treated with non-removable devices, the tooth enamel is subjected to additional mechanical impact of orthodontic forces locally. Numerous methods are used for the prevention of caries using various organic and inorganic fluorine compounds: sodium fluoride and potassium; stannous fluoride; amino fluoride; monofluorophosphate; zirconium fluoride; and others [14–18]. One of these methods is deep fluorination, which is based on the use of new-generation fluorine-containing solutions that promote the formation of microcrystalline CaF₂, which releases fluorine for a long time (more than 1 year) [19–21]. Effective prevention, diagnosis, and treatment of lesions will minimize caries formation and also provide an aesthetic smile. The most appropriate, cheapest, and easiest-to-apply treatment method should be preferred for the patient.

Despite the large number of studies devoted to examining the effect of non-removable orthodontic devices on changes in the tissues of the dentition, there is no information on the dynamics of the state of the ultrastructure of the hard tissues of the tooth at the stages of orthodontic treatment and related complications. Split-mouth randomized controlled trials (RCTs) are popular in oral health research. Split-mouth CTs have the advantage that most of the variability among patients is excluded from the assessment of the effect of the intervention to potentially increase statistical power, with each subject being its own control [22–25]. The purpose of this research is determined by the necessity to study the state of hard tissues of the teeth and carry out rational preventive measures to reduce the risk of complications in the process of orthodontic treatment using a non-removable device.

2. Materials and Methods

2.1. Study Design

The studies were carried out at the Dental Academy clinic and at the Department of Pediatric Dentistry and Orthodontics of YSMU. The scientific work consisted of clinical and laboratory studies. For the in vitro *study*, 68 teeth of different groups were used (mainly the first or second premolars of the upper or lower jaws, or the central and lateral incisors in the jaw, Figure 1). These teeth were removed according to orthodontic instructions. *The clinical study* was conducted with 59 orthodontic patients aged 12–17 who were treated with non-removable orthodontic appliances.



Figure 1. Representation of the distribution of groups and subgroups of teeth in this study.

2.2. In Vitro Study

For the laboratory study, the extracted human teeth were used as a model for simulating ex vivo procedures. Human tissues are the first choice of material for evaluating the bond strength. To obtain comparable results, the samples were stored under certain uniform conditions. The antimicrobial properties of the preservation medium are rather critical as extracted teeth present a risk of cross-infection. The sterilization process should be neutral for the microstructure of enamel and dentine, as even a slight change can affect the adhesive bond [26,27]. The extracted teeth were directly stored in a 0.1% thymol solution and then each tooth was treated with 0.12% chlorhexidine for disinfection. Residual soft tissues were removed with a surgical lancet. The enamel surface of all the teeth was treated with a polishing abrasive (not containing fluorine) and distilled water, using circular brushes at a slow speed.

Next, the teeth were washed with distilled water and randomly divided into two equal groups (A and B). A small window was cut out of the adhesive tape according to the size of the bracket base. The tape was superimposed on the vestibular surface of the teeth and the window was located around the bracket. This was carried out to isolate tooth enamel from the acid treatment to avoid unintended demineralization. The part of the enamel that was within the adhesive tape was treated with 30% phosphoric acid for 30 s. After this process, the enamel was washed with distilled water for the same amount of time and thoroughly dried with a stream of air. After these procedures, metal braces manufactured by the Italian company Leone (Florence, Italy) were installed on all the teeth using special Brace Paste[®] Adhesive (American Orthodontics, Sheboygan, WI, USA).

After the installation of braces, the teeth of the first group A were not subjected to prophylactic treatment. The teeth of the second group B were treated with an enamel-sealing liquid Tiefenfluorid (Humanchemie GmbH, Alfeld, Germany) [20,21,28], with 2 mm around the base of the bracket. Then, the teeth of each group were placed in separate containers that contained a solution of artificial saliva (20 μ mol/L NaHCO₃, 3 μ mol/L NaH₂PO₄, and 1 μ mol/L CaCl₂ under neutral pH conditions).

2.3. Artificial Caries Induction

After 12 h, the teeth were twice placed for 1 h into 200 mL of artificially demineralizing (caries-stimulating) solution (2.2 μ mol/L Ca²⁺, 2.2 μ mol/L PO₄, 50 μ mol/L acetic acid under conditions pH 4.4) with an interval of 6 h. Both solutions were kept in an incubator at a temperature of 37 °C corresponding to normal oral temperature and were renewed every 3 days during the 35 days of the study. All the teeth were removed from the solution, rinsed

with deionized water for 5 s, and then brushed without the use of abrasives. Thus, the teeth of both groups were brushed twice a day. The teeth were transferred into two solutions (an artificial saliva solution and a caries-stimulating solution), adhering to the above rules for all 35 days. The teeth of the second group B on the 15th day were additionally treated with Tiefenfluorid. The transfer to solutions was carried out to obtain similar conditions in the clinical environment. After 35 days, all the teeth were removed from the solutions and dried; the brackets were removed, washed with distilled water, and then stored in boxes with distilled water.

2.4. Morphological Study

The preparation of slides for microscopic examination was carried out in the following order:

- 1. Decalcification for 3 days in 5% nitric acid solution.
- 2. Washing with running tap water during the day 5–6 times to remove the acid solution.
- 3. Washed tissues were transferred to alcohol solutions of increasing concentrations for the dehydration procedure.
- 4. After dehydration, a cleaning procedure was carried out by soaking in a xylol solution.
- 5. The cleaned tissues were transferred into soft paraffin (melting point 46–48 °C).
- 6. Then they were blocked by pouring into hard paraffin (melting point 56–58 °C).
- 7. Preparation of 5 μm, 7 μm, and 10 μm thick sagittal sections taken from blocked tissues by means of a rotary microtome (Leitz, Wetzlar, Germany).
- 8. Staining with Ehrlich's hematoxylin for 5–10 min.
- 9. After staining, the sections were dehydrated and covered with a coverslip [26].

The preparations were examined under the NIKON Eclipse Ci-L electron microscope (Nikon, Japan) at $\times 100$ or $\times 200$ magnification. In the part being examined in the area of the bracket, the depth and surface area of the demineralized enamel layer were measured. All examinations were performed using the same microscope to avoid possible differences between instruments. The average indicators for each group were calculated and a comparative analysis was carried out between the indicators of the groups.

2.5. Exclusion Criteria of Clinical Study

Exclusion criteria were used only in the clinical study, there were several:

- The presence of white spots on the teeth before fixing the braces.
- Refusal of the patient or parents to participate in the study.
- Ineligibility for the studied age groups.

2.6. In Vivo Study

Participation in this study was voluntary. All participants signed a consent form in accordance with the Declaration of Helsinki. The split-mouth technique [22,23] was used. Immediately after fixing the brackets, the areas around the base of the bracket on the teeth of one-half of the oral cavity were treated with an enamel-sealing liquid, Tiefenfluorid [21,29]. The teeth of the second half of the oral cavity served as a control group. The treatment was carried out every 6 months during the entire treatment period. The presence of a white spot on the surfaces of the teeth was determined. The demineralization and remineralization processes of WSL were studied using laser-fluorescence DIAGNOdent [12,13,30].

2.7. Statistical Analysis

Descriptive analyses (mean \pm SD for continuous variables and frequencies/proportion for categorical variables) were computed for all variables of interest. Differences between the two groups were evaluated using the "T-Test" for categorical variables. *p*-value was considered significant at <0.05 and <0.001 for highly significant results. Analyses were conducted using the SPSS19 (SPSS Inc., Chicago, IL, USA) software.

3. Results

3.1. Results of Studies In Vitro

By the end of the study, tooth loss was observed in both study groups: two each in the A (control) and B (prophylactic) groups. The loss is explained by the damage they suffered during the manufacture of microscopic preparations. In each group, 32 teeth remained. In the control group, the proportion of premolars was 81.25%, and for incisors, it was 18.75%. In the prophylactic group, the proportion of premolars was 75%, and for incisors, 25%.

The average depths of decalcification in the control and prophylactic groups are shown in Figure 2. The results obtained are statistically significant: t = 11.7, p < 0.0001 in the control group and t = 10.02, p < 0.0001 in the prophylactic group.



Figure 2. Depth of enamel damage in the control and prophylactic groups in microns. * The results obtained are statistically significant: p < 0.0001.

The area of the lesion in the groups was also different (Figure 3) and statistically significant: t = 17.2, p < 0.0001 in the control group and t = 11.4, p < 0.0001 in the prophylactic group.



Figure 3. Area of enamel lesion in the staining stage in the control and prophylactic groups in mm^2 . * The results obtained are statistically significant: p < 0.0001.

3.2. Results of Morphological Study

When discussing the study results, the obtained sections showed a statistically significant difference between the prophylactic and control groups of teeth in terms of the depth of the lesion and the surface area (t = 5.3, p < 0.0001 and t = 2.4, p < 0.02, respectively). Thus, decalcification is more pronounced in the first (control group), where the average thickness of the affected enamel layer is 1.83 times greater than in the first one (Figures 4 and 5).



Figure 4. The focus of deep demineralization of tooth enamel in the control group (microscopic examination of the enamel in the area of the carious spot revealed changes in the crystals: a violation of the orientation of the crystals in the structure of hydroxyapatites, a change in the shape of the crystals, their size, and the appearance of crystals atypical for normal enamel. Staining with Ehrlich's hematoxylin, in magnification $\times 200$).



Figure 5. Superficial enamel lesion in the control group (during microscopic examination of enamel, changes develop only in the surface layers of enamel in the form of displacement of enamel prisms relative to each other. Staining with Ehrlich's hematoxylin, in magnification \times 200).

7 of 10

3.3. Clinical Results

The study involved patients aged 12 to 17 years who were divided into two groups: 12–14 years (32 patients, group C—control, and group D—preventive), and 15–17 years (27 patients, group E—control, and group F—prophylactic).

According to the preliminary data, in both groups, deterioration was observed during the period of orthodontic treatment. In both groups, 34 patients (57.6%), 176 teeth (21.6%), had demineralization phenomena in the form of white spots. A comparative analysis revealed a certain difference in the degree of damage in the control and prophylactic groups in favor of the prophylactic group, where the statistical significance was $p \leq 0.01$.

In the group with patients aged 12–14 years (mean age 12.88 \pm 0.83), white spots formed in 20 patients (62.5%) on 112 teeth (23.3%). On average, 3.5 ± 3.2 teeth were affected in one patient in this group. In the control group, white spots were detected on 68 teeth (60.7%). There were more white spots on the anterior teeth—49 (72.1%) and fewer on the posterior teeth of the lateral group—19 (27.9%) (t = 5.2; $p \le 0.0001$). In the prophylactic group, 44 teeth (37.7%) were affected, of which 31 were incisors (70.5%) and 13 were molars (29.5%) (t = 3.48; $p \le 0.002$). The difference between the identified foci of demineralization in the control and prophylactic groups of the patients aged 12–14 years was statistically significant (t = 4.98; $p \le 0.001$). In the group of patients aged 15–17 years (mean age 15.96 \pm 0.8), white spots were detected in 14 people (51.9%) on 64 teeth (19.05%). On average, 2.37 ± 2.8 teeth were affected in one patient in this group. In the control group, white spots were detected on 41 teeth (64.06%). Likewise, more on the incisors—29 (70.7%), and fewer on the posterior teeth of the lateral group—12 (29.3%) (t = 3.25; p = 0.003). In the prophylactic group, 23 teeth (35.9%) were affected, of which 13 were incisors (56.5%) and 10 were molars (43.5%) (t = 1.1; p = 0.265). The difference between the detected foci of demineralization in the control and prophylactic groups of the patients aged 15–17 years was statistically significant (t = 2.449; p = 0.021). In the control groups (C, E), no white spots were found upon initial examination. Upon final examination, the number of spots was 109 (61.9%). In the prophylactic groups (D and F) over the same time period, the number of white spots was 67 (38.1%) (t = 5.3; $p \le 0.001$). As a result of observation, the number of teeth with white spots in children aged 12–14 years was more (63.6%) than the index of the age group of 15–17 years (36.4%) in all the mentioned groups (C, D, E, and F).

When analyzing these indicators of laser fluorometry, the following results were obtained. Before the start of orthodontic treatment in the group of 12–14-year-old patients, the mean values of KaVo DIAGNOdent were 9.69 ± 2.48 ($p \le 0.0001$). After 6 months of the orthodontic treatment and preventive measures with Tiefenfluorid, in the prophylactic group, the average fluorometry values increased to 13.66 ± 3.67 ($p \le 0.0001$). In the control group, the average fluorometry values increased to $15.75 \pm 5.4 p$ ($p \le 0.0001$). In patients with white spots in group D, the indices were 16 ± 2.34 , and in group C, they were 19.45 ± 2.98 . The difference in indicators was statistically significant (t = 12.96; $p \le 0.0001$).

Before the startup of orthodontic treatment in the group of 15–17-year-old patients, the mean values of KaVo DIAGNOdent were 13.52 ± 2.48 ($p \le 0.0001$). After 6 months of the orthodontic treatment and prophylactic measures with Tiefenfluorid, the mean fluorometry values increased to 13.33 ± 4.3 in the prophylactic group and to 14.67 ± 5.05 in the control group ($p \le 0.0001$). In the patients with white spots in group F, the indices were 16.79 ± 3.04 , and in group E, the indices were 18.93 ± 2.79 . The difference in indicators was statistically significant (t = 6.51; $p \le 0.0001$).

4. Discussion

The analysis of the obtained results of the experiment confirmed parallel clinical observations of the same patients using the split-mouth technique. This created equal conditions for both groups, and ensured statistical significance. The indicators of the prevalence of white spots in patients receiving orthodontic treatment were different. In patients in the 12–14-year-old age group, the foci of demineralization were found more

often (58.8%). This can also be explained by the incomplete mineralization of tooth enamel at this age [9,10,31].

Non-removable orthodontic appliances create additional points for the accumulation of food debris, increasing the surface area to which microbial biofilm can attach. Due to their irregular shape, it is almost impossible to completely remove microbial biofilm. As a result, one can infer that orthodontic treatment is a risk factor for developing white spots. As a rule, white lesions due to orthodontic treatment occur on the buccal surface of the teeth, and are difficult to clean. This is the area between the brackets and the gum, especially in the upper lateral incisors and canines, as well as in the lower canines and first premolars. This is supported by the data of some investigators who noted the development of white spots in 15 to 60% of cases [8,32]. These lesions can develop very rapidly over 4 weeks, affecting deeper areas, making them impossible to re-calcify. Therefore, it is very important to identify these lesions as quickly as possible [14]. Unfortunately, once these lesions become clinically visible, the changes become irreversible.

Clinical studies have confirmed that the regular use of a prophylactic agent, which does not depend on patient compliance during the orthodontic treatment, reduced the risk of a cariogenic situation by 38.07%. Protection of the tooth surface during orthodontic treatment is significant, considering the high incidence of enamel mineral loss that occurs during non-removable appliance treatment at aesthetically important sites, such as the vestibular surfaces of maxillary incisors [4,29,33]. The study showed that in 69.3% of cases, the lesions were localized on the anterior teeth. The prevalence of foci of demineralization on molars was 30.7%, which can be explained by the peculiarity of fixing braces on this group of teeth (the use of bondage rings, which occupy a significant surface area and so accumulate even more microbial biofilm on their surface). In addition, the patients face difficulties in brushing this group of teeth due to their location [3].

An in vitro study showed that the use of a prophylactic agent around the base of the bracket leads to a decrease in the depth of enamel demineralization by 1.8 times, confirming the hypothesis of the study. The decrease in depth is predetermined by two-factor action: Tiefenfluorid in combination with highly dispersed calcium hydroxide realizes deep fluoridation of tooth enamel by the formation of submicroscopic CaF₂ crystals. Packed in a gel and thus protected from mechanical influences, these crystals constantly emit fluorine, which, in turn, together with the mineral salts of saliva provide long-term remineralization, enhancing the fluorine by almost 100 times [19,20]. The protective properties of fluorine, which have been absorbed onto the surface of the enamel, disappear over time. This is why coating with Tiefenfluorid was carried out every 6 months.

During the study of histological sections, enamel decalcification was detected. There is a difference between the thicknesses of the layers, which indicates more pronounced demineralization processes in one group rather than in the other. Since the selection of teeth for groups was random, this difference cannot be a hidden or obvious reason for demineralization ($p \le 0.001$). A limitation of the study is that, in the experimental groups, the extracted teeth under examination were not located in their natural medium; there was no natural nutrition of the teeth. This limitation was compensated by the circumstances in which the experimental group was compared with the control group, where the teeth were processed by the same methods, in the most approximate conditions. In vitro studies allow one to study sections of teeth, which enable factual description of the formation of demineralization, as well as measuring the depth of the affected area.

5. Conclusions

The obtained data and evaluation of the local fluoridation method using the enamelsealing liquid Tiefenfluorid for deep fluoridation determined that Tiefenfluorid has powerful bactericidal activity and helps to suppress the formation of a microbial biofilm on the enamel surface. This is due to the inclusion of copper ions in the composition, which serve as an incentive to develop alternative methods for increasing the resistance of enamel. Thus, further studies of this new and promising solution are needed. Author Contributions: Conceptualization, M.M. (Mikayel Manrikyan); data curation, G.K. and G.M.; formal analysis, I.V.; investigation, G.K. and G.M.; methodology, G.K. and G.M.; software, I.V.; supervision, M.M. (Marina Markaryan); validation, M.M. (Marina Markaryan); writing—original draft, G.K., M.M. (Mikayel Manrikyan) and G.M.; writing—review and editing, M.M. (Marina Markaryan) and I.V. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: The study was conducted in accordance with the Declaration of Helsinki and approved by the Ethics Committee of Yerevan State Medical University (Approval No. 2 dated 18 October 2018).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Not available.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Bonetti, A.G.; Zanarini, M.; Incerti Parenti, S.; Lattuca, M.; Marchionni, S.; Gatto, M.R. Evaluation of enamel surfaces after bracket debonding: An in-vivo study with scanning electron microscopy. *Am. J. Orthod. Dentofac. Orthop.* 2011, 140, 696–702. [CrossRef] [PubMed]
- Bora, N.; Mahanta, P.; Kalita, D.; Deka, S.; Konwar, R.; Phukan, C. Enamel Surface Damage following Debonding of Ceramic Brackets: A Hospital-Based Study. Sci. World J. 2021, 2021, 5561040. [CrossRef]
- 3. Kozak, U.; Sękowska, A.; Chałas, R. The Effect of Regime Oral-Hygiene Intervention on the Incidence of New White Spot Lesions in Teenagers Treated with Fixed Orthodontic Appliances. *Int. J. Environ. Res. Public Health* **2020**, *17*, 9460. [CrossRef] [PubMed]
- 4. Akay, E.N. Preventive Methods and Treatments of White Spot Lesions in Orthodontics. In Oral Health Care—An Important Issue of the Modern Society; Ardelean, L., Rusu, L.C., Eds.; Intech Open: London, UK, 2022. [CrossRef]
- 5. Pont, H.B.; Özcan, M.; Bagis, B.; Ren, Y. Loss of surface enamel after bracket debonding: An in-vivo and ex-vivo evaluation. *Am. J. Orthod. Dentofac. Orthop.* **2010**, *138*, 387.E1–387.E9. [CrossRef]
- Sarafopoulou, S.; Zafeiriadis, A.A.; Tsolakis, A.I. Enamel Defects During Orthodontic Treatment. Balk. J. Dent. Med. 2018, 22, 64–73. [CrossRef]
- 7. Gittner, R.; Müller-Hartwich, R.; Engel, S.; Jost-Brinkmann, P.G. Shear bond strength and enamel fracture behavior of ceramic brackets fascination[®] and fascination[®] 2. *J. Orofac. Orthop.* **2012**, *73*, 49–57. [CrossRef]
- 8. Khoroushi, M.; Kachuie, M. Prevention and Treatment of White Spot Lesions in Orthodontic Patients. *Contemp. Clin. Dent.* **2017**, *8*, 11–19. [CrossRef]
- 9. Metin-Gürsoy, G.; Uzuner, F.D. The Relationship between Orthodontic Treatment and Dental Caries. In *Dental Caries—Diagnosis, Prevention and Management*; Intech Open: London, UK, 2018. [CrossRef]
- 10. Zou, J.; Meng, M.; Law, C.S.; Rao, Y.; Zhou, X. Common dental diseases in children and malocclusion. *Int. J. Oral Sci.* 2018, 10, 7. [CrossRef]
- 11. Heymann, G.C.; Grauer, D. A contemporary review of white spot lesions in orthodontics. J. Esthet. Restor. Dent. 2013, 25, 85–95. [CrossRef]
- Carvalho, F.B.; Barbosa, A.F.; Zanin, F.A.; Brugnera Júnior, A.; Silveira Júnior, L.; Pinheiro, A. Use of laser fluorescence in dental caries diagnosis: A fluorescence x biomolecular vibrational spectroscopic comparative study. *Braz. Dent. J.* 2013, 24, 59–63. [CrossRef]
- Kavvadia, K.; Seremidi, K.; Reppa, C.; Makou, M.; Lagouvardos, P. Validation of fluorescence devices for evaluation of white spot lesions in orthodontic patients. *Eur. Arch. Paediatr. Dent.* 2018, 19, 83–89. [CrossRef] [PubMed]
- 14. Nascimento, P.L.; Fernandes, M.T.; Figueiredo, F.E.; Faria-E-Silva, A.L. Fluoride-Releasing Materials to Prevent White Spot Lesions around Orthodontic Brackets: A Systematic Review. *Braz. Dent. J.* **2016**, *27*, 101–107. [CrossRef] [PubMed]
- Mazur, M.; Jedliński, M.; Ndokaj, A.; Ardan, R.; Janiszewska-Olszowska, J.; Nardi, G.M.; Ottolenghi, L.; Guerra, F. Long-Term Effectiveness of Treating Dentin Hypersensitivity with Bifuorid 10 and Futurabond U: A Split-Mouth Randomized Double-Blind Clinical Trial. J. Clin. Med. 2021, 10, 2085. [CrossRef] [PubMed]
- 16. Greenwall, L. White lesion eradication using resin infiltration. Int. Dent. Afr. Ed. 2013, 3, 54–62.
- 17. Devji, T. Silver diamine fluoride is probably more effective than atraumatic restorative treatment, fluoride varnish, or no treatment for controlling caries progression in children. *J. Am. Dent. Assoc.* **2018**, 149, e65. [CrossRef]
- 18. Kooshki, F.; Pajoohan, S.; Kamareh, S. Effects of treatment with three types of varnish remineralizing agents on the microhardness of demineralized enamel surface. *J. Clin. Exp. Dent.* **2019**, *11*, e630–e635. [CrossRef]
- 19. Meto, A.; Meto, A.; Tragaj, E.; Lipo, M.; Bauermann, C. The Use of Tiefenfluorid for Desensitization of Dentinal Hyperesthesia. *Balk. J. Dent. Med.* **2014**, *18*, 85–88. [CrossRef]
- Kozaczuk, S. Deep Penetration Fluoridation for Caries Prevention and Treatment: The Use of Tiefenfluorid[®] Junior in Children. Case Reports; Nowa Stomatologia: Kraków, Poland, 2020; pp. 15–25. [CrossRef]

- 21. Knappvost, A. Indications for Use and Mechanisms of Cares Prophylactic Action of Preparations of Deep Fluoridation—Enamel-Sealing and Dentin-Sealing Liquids; Dentistry for All: West New York, NJ, USA, 2001; pp. 38–42.
- Femiano, F.; Femiano, R.; Femiano, L.; Nucci, L.; Minervini, G.; Antonelli, A.; Bennardo, F.; Barone, S.; Scotti, N.; Sorice, V.; et al. A New Combined Protocol to Treat the Dentin Hypersensitivity Associated with Non-Carious Cervical Lesions: A Randomized Controlled Trial. *Appl. Sci.* 2021, 11, 187. [CrossRef]
- 23. Zhu, H.; Zhang, S.; Ahn, C. Sample size considerations for split-mouth design. *Stat. Methods Med. Res.* 2017, 26, 2543–2551, Erratum in *Stat. Methods Med. Res.* 2019, 28, 3487. [CrossRef]
- 24. Smaïl-Faugeron, V.; Fron-Chabouis, H.; Courson, F.; Durieux, P. Comparison of intervention effects in split-mouth and parallel-arm randomized controlled trials: A meta-epidemiological study. *BMC Med. Res Methodol.* **2014**, *14*, 64. [CrossRef]
- 25. Pozos-Guillén, A.; Chavarría-Bolaños, D.; Garrocho-Rangel, A. Split-mouth design in Paediatric Dentistry clinical trials. *Eur. J. Paediatr. Dent.* 2017, *18*, 61–65. [PubMed]
- Lee, J.J.; Nettey-Marbell, A.; Cook, A., Jr.; Pimenta, L.A.; Leonard, R.; Ritter, A.V. Using extracted teeth for research: The effect of storage medium and sterilization on dentin bond strengths. J. Am. Dent. Assoc. 2007, 138, 1599–1603. [CrossRef] [PubMed]
- Nawrocka, A.; Łukomska-Szymańska, M. Extracted human teeth and their utility in dental research. Recommendations on proper preservation: A literature review. *Dent. Med. Probl.* 2019, 56, 185–190. [CrossRef] [PubMed]
- Qeli, E.; Toti, Ç.; Odorici, A.; Blasi, E.; Tragaj, E.; Tepedino, M.; Masedu, F.; Kaçani, G.; Hysi, D.; Meto, A.; et al. Effectiveness of Two Different Fluoride-Based Agents in the Treatment of Dentin Hypersensitivity: A Prospective Clinical Trial. *Materials* 2022, 15, 1266. [CrossRef]
- 29. Rogers, S.; Chadwick, B.; Treasure, E. Fluoride-containing orthodontic adhesives and decalcification in patients with fixed appliances: A systematic review. *Am. J. Orthod. Dentofac. Orthop.* **2010**, *138*, 390.E1–390.E8. [CrossRef]
- Salmerón-Valdés, E.; Lara-Carrillo, E.; Medina-Solís, C.; Robles-Bermeo, N.L.; Scougall-Vilchis, R.J.; Casanova-Rosado, J.F.; Pontigo-Loyola, A.P.; Fernández Barrera, M.Á. Tooth demineralization and associated factors in patients on fixed orthodontic treatment. *Sci. Rep.* 2016, *6*, 36383. [CrossRef]
- 31. Choi, Y.Y. Relationship between orthodontic treatment and dental caries: Results from a national survey. *Int. Dent. J.* **2020**, 70, 38–44. [CrossRef]
- Srivastava, K.; Tikku, T.; Khanna, R.; Sachan, K. Risk factors and management of white spot lesions in orthodontics. J. Orthod. Sci. 2013, 2, 43–49. [CrossRef]
- Zarzycka-Kogut, K.; Pucek, M.; Jolanta, S. Orthodontic Treatment—Complications and Preventive Measures. *Pol. J. Public Health* 2014, 124, 103–106. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.